Author's response to reviews

Title: An analysis of risk factors of non-fatal drowning among children in rural areas of Guangdong Province, China: A Case-Control Study

Authors:

Wenjun Ma (mwj68@tom.com)
Shaoping Nie (Nieshp@163.com)
Haofeng Xu (xhf62@163.com)
Yanjun Xu (gdxyj05@21cn.com)
Yurun Zhang (gdncd@126.com)
Xiuling Song (songxl918@163.com)

Version: 4 Date: 27 February 2010

Author's response to reviews:

Dear editor

I have reversed reference 41 and 42 in the new manuscript, and made some revision to response to the reviewer's comments.